Author's response to reviews

Title: Aortitis requiring aortic repair associated with glaucoma, thyroiditis, glaucoma, and neuropathy: Case report

Authors:

Claudia Stollberger (claudia.stoellberger@chello.at)
Marion Avanzini (marion.avanzini@wienkav.at)
Aileen Hanafin (aileen.hanafin@doctors.org.uk)
Ramona Sanani (ramona.sanani@wienkav.at)
Georg Wieselthaler (georg.wieselthaler@wienkav.at)
Nikolaus Wick (nikolaus.wick@wienkav.at)
Günther Bayer (guenther.bayer@wienkav.at)
Günther Mölzer (guenther.moelzer@wienkav.at)
Josef Finsterer (josef.finsterer@wienkav.at)

Version: 3 Date: 7 May 2011

Author's response to reviews:

Dear Editor,

Following changes have been made:

Title: please ensure that the title is identical in the manuscript and the submission system - DONE

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs) - DONE

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution. - DONE

Abstract: This should be identical in the manuscript and the submission system. - DONE

Authors Contributions: Please put this section in paragraph form. - DONE

Abbreviations: Please put this section into paragraph form. - DONE
Figures: Please provide figure titles. These should be concise and be the first sentence of the figure legend. - DONE

Minor revisions

Email address: please include email addresses for all the authors on the title page, using the following format: (...) - NOT DONE

Thank you for your comments.

Best regards

Claudia Stöllberger, Corresponding author